Author’s response to reviews

Title: The Alu-insertion progesterone receptor gene polymorphism is not associated with breast cancer: a meta-analysis

Authors:

Jun Yao (yaojun198717@163.com)
Xing-ling Qi (995140327@qq.com)
Yong Zhang (zhangyong@cancerhosp-ln-cmu.com)

Version: 2 Date: 11 Jan 2018

Author’s response to reviews:

Response to Editor Comments:

Comment 1. There appears to be discrepancy in the new Figure 1. Should have N = 10 studies (have "N = 9, 10 case-control studies").

Answer. We have revised the description in the new Figure 1. Please check it.